Topographic effect creates non-climatic variations in ice-core based temperature records of the last millennium

Remi Dallmayr¹, Johannes Freitag², Thomas Laepple², Frank Wilhelms³, Daniela Jansen², Melanie Behrens¹, and Maria Winona Hoerhold²

¹Alfred Wegener Institut for Polar and Marine Reasearch ²Alfred Wegener Institute for Polar and Marine Research ³Alfred Wegener Institute for Polar and Marine Sciences

November 24, 2022

Abstract

Past temperature reconstructions from polar ice sheets are commonly based on stable water isotope records in ice-cores. However, despite major efforts in the understanding of the ice-core signal formation, the temperature reconstructions of the last millennium in Antarctica remain highly uncertain. Here, using a 100 km scale representative surface water isotope dataset, we show that the spatial variability of local surface topography and accumulation rate anomalies influences the isotopic composition of the upper-meter snowpack. The magnitude of this non-temperature effect on water isotopes is similar to changes of the last millennium. We demonstrate that these spatial anomalies are advected into the deeper firn and ice column, and are translated into an artificial centennial to millennial scale variability in the isotope record. Additionally, we provide an estimation of areas where this effect is relevant for last millennium temperature reconstructions.

Hosted file

supplement_information_topographic_effect_creates_non-climatic_variations_in_ice-core_based_temperature
available at https://authorea.com/users/550176/articles/603893-topographic-effect-createsnon-climatic-variations-in-ice-core-based-temperature-records-of-the-last-millennium

Hosted file

essoar.10508113.1.docx available at https://authorea.com/users/550176/articles/603893topographic-effect-creates-non-climatic-variations-in-ice-core-based-temperaturerecords-of-the-last-millennium

Topographic effect creates non-climatic variations in ice-core based temperature records of the last millennium

Remi Dallmayr^{1,4}, Johannes Freitag¹, Thomas Laepple^{2,3}, Frank Wilhelms^{1,4}, Daniela Jansen¹, Melanie Behrens¹ and Maria Hörhold¹

¹ Alfred-Wegener-Institut Helmholtz-Zentrum für Polar-und Meeresforschung, Bremerhaven, Am Handelshafen 12, 27570 Bremerhaven, Germany

²Alfred-Wegener-Institut Helmholtz-Zentrum für Polar-und Meeresforschung, Potsdam, Telegrafenberg A45, 14473 Potsdam, Germany

³University of Bremen, MARUM – Center for Marine Environmental Sciences and Faculty of Geosciences, 28334 Bremen, Germany

⁴GZG Abt. Kristallographie, University of Göttingen, Göttingen, Germany

 $Corresponding \ author: \ remi.dallmayr@awi.de$

Key points:

- The isotopic composition in surface snow is anti-correlated to local accumulation rate anomalies, driven by the local surface topography.
- Spatial variability is advected into an ice-core site by ice-flow, inducing millennial scale non-climatic variation in the isotopic record.
- Map of time scales variations due to topography effect in East Antarctica for consideration in the interpretation of records from ice-cores.

Key words:

Antarctic isotope record, surface topography, accumulation rate, spatial and temporal variability, paleoclimate reconstructions

Abstract

Past temperature reconstructions from polar ice sheets are commonly based on stable water isotope records in ice-cores. However, despite major efforts in the understanding of the ice-core signal formation, the temperature reconstructions of the last millennium in Antarctica remain highly uncertain. Here, using a 100 km scale representative surface water isotope dataset, we show that the spatial variability of local surface topography and accumulation rate anomalies influences the isotopic composition of the upper-meter snowpack. The magnitude of this non-temperature effect on water isotopes is similar to changes of the last millennium. We demonstrate that these spatial anomalies are advected into the deeper firn and ice column, and are translated into an artificial centennial to millennial scale variability in the isotope record. Additionally, we provide an estimation of areas where this effect is relevant for last millennium temperature reconstructions.

Plain Language Summary

Stable water isotopes measured in ice-cores are commonly used as a paleothermometer. However, these records can be affected by multiple processes and their reliability is unclear, particularly for centennial to millennial time scales in Antarctica. We demonstrate that in low accumulation rate regions, the local topography can create significant spatial variability in water isotopes. This can create temporal variability in ice-core records as these anomalies are advected into the core by ice-flow. This mechanism creating non-temperature related water isotope variations on time-scales from centuries to millennia is likely omnipresent within the continent, potentially affecting numerous ice-cores. We estimate relevant areas with respect to this effect to assist future coring efforts, as well as the careful interpretation of existing records.

1. Background and motivation

Stable water isotopes in ice are commonly used as a proxy for paleotemperatures (Jouzel et al., 1997) and allow reconstructions of past temperature from ice-cores (PAGES 2k Consortium, 2013, Steig et al., 2013). Despite major advances in the analysis methods and field studies to understand signal formation of the isotope signal preserved in the firn and ice (Casado et al., 2018; Münch et al., 2017; Münch & Laepple, 2018), temperature reconstructions from East-Antarctic ice cores are still highly uncertain on centennial to millennial scale and often show a low correlation with observations and simulations (Jones et al., 2016; Klein et al., 2017).

Stratigraphic noise (Fischer et al., 1985, Münch et al., 2017) masks climate signals from interannual to multi-decadal scales but can be minimized by averaging multiple records or taking temporal averages (Münch et al., 2016). Post-depositional effects near the surface might affect the isotopic composition (Neumann & Waddington, 2004; Town et al., 2008) but their importance for ice-core records is under discussion (Münch et al., 2017). Precipitation intermittency acts on longer spatial scales and leads to strong noise in water isotope records but is generally well understood and averages out on long-timescales (Casado et al., 2020).

While these processes create a strong non-climatic variability, they cannot explain observed differences in the nearby ice-core isotope records that can last for several centuries. One example is the ice-core site B31 in Dronning Maud Land, with a drop in stable water isotopic composition accompanied by a sudden increase in accumulation rate (Oerter et al., 2004). The nearby core B32 does not confirm these changes. Radar-data within the ~100 km distance between sites show strong local variations of accumulation rate within short distances (up to ~30 kg m⁻² a⁻¹, i.e. ~50%, Rotschky et al., 2004). Precipitation varying only slightly on such scale, these variations were related to surface undulations caused by bedrock topography (Eisen et al., 2004; Rotschky et al., 2004), providing a first hint to a relation of topography, accumulation and isotopic composition.

A relationship between these variables has been suggested by previous studies. First, surface undulations were found to lead to spatial changes in deposition of snow (Arcone et al., 2005; Black & Budd, 1964; Eisen et al., 2005; Frezzoti et al., 2002a, 2004; Fujita et al., 2011; Rotschky et al., 2004, 2007; Whillans, 1975). Second, Ekaykin et al. (2016) observed significant variations of the upper-meter isotopic composition along the slopes of a surface undulation within a megadune area near Vostok station. They proposed that the spatial variability can lead to temporal non-climatic variations of the record (Ekaykin et al., 2016), due to the upwind drift of the dune (Frezzotti et al., 2002a). This situation is limited to particular areas covering only small parts of the Antarctica ice sheet (Fahnestock et al., 2000), but raises the question to what extend variations in paleo-climate records may be biased by local topography-induced variations in accumulation rate.

Motivated by these findings we designed a specific study to evaluate the effect of surface topography on the stable-water-isotopic composition. Strongly replicated sampling of the top meter of snow, together with Ground Penetrating Radar (GPR) and Global Positioning System (GPS) data on a 115 km traverse including the B31 site allowed us for the first time a representative characterization of the near-surface isotopic composition on a large-scale and its relationship with the local topography.

2. Methods and data



Figure 1. The area of investigation (left) and the field work results (right).

Left panel: Traverses -00/01 (red line) and -18/19 (this study, black line) in between the B31 and B32 sites (white circles). The sampling sites of season 18/19 are indicated with yellow crosses. The topography is shown as 1m elevation contours (Howat et al., 2019). Right panel: a) The elevation along the traverse-18/19 (black line) with yellow points marking the sampling sites. b) Accumulation rate along the traverse-00/01 (red line, Rotschky et al., 2004). c) 18 O at sites, (blue markers connected by dashed line) with the error-bars indicating the estimated remaining uncertainty (1 standard-error) from stratigraphic noise. The solid gray line represents the linear trend of 18 O versus elevation.

In the austral field season 2018/2019 we performed a 115 km long traverse (hereafter traverse-18/19) in the interior of Dronning Maud Land, East Antarctica (Fig. 1) connecting the B31 ice-core site ($3^{\circ}25.82$ 'W 75°34.890 S, 2680 m.a.s.l.) to the B32 ice-core site ($0^{\circ}00.42$ 'E 75°00.14 'S, 2892 m.a.s.l) and Kohnen station (EDML deep ice-core site, $0^{\circ}04E$ 75°00 S, 2892 m.a.s.l.).

A post-processed kinematic (PPK) global positioning system mounted at the living container sampled geodetic positions at 1 Hz (i.e. \sim 3.5 m), and allowed to derive the surface elevation profile of the traverse (Fig. 1).

We further make use of accumulation data based on GPS and GPR measurements (Eisen et al., 2004; Rotschky et al., 2004) from a traverse in 2000/2001 (hereafter traverse-00/01) to estimate the surface mass-balance profile (Fig. 1). For this, we use the internal radar reflection horizon corresponding to AD1817 \pm 5 a (i.e. \pm 2.5% on accumulation rate) (Rotschky et al. 2004).

2.1 Stable water isotopes sampling positions and strategy

Along the route-18/19, 27 sampling sites were chosen with regard to the decorrelation length of the topographic anomalies, i.e. 2.5 km distance between sites for the first 15 km of the traverse, and 5 km later on (Fig. 1). With the knowledge gained from trench studies (Laepple et al., 2016; Münch et al., 2016) and the use of a newly developed fast sampling tool (Dallmayr et al., 2020), we extracted and averaged at each site 10 upper-meter samples with 10 m distance to minimize the uncertainty from stratigraphic noise (Fisher et al., 1985). The mean value of uncertainty is estimated as $0.3 \%_0$, (1 standard-error) from the remaining variability of the replicated one-meter averages. (see Text S1 and Table S1, supporting information).

2.2 Measurement of stable water isotopes

All samples were kept frozen during shipment and storage until measurement at the Alfred-Wegener-Institut Helmholtz-Zentrum für Polar-und Meeresforschung (AWI) in Bremerhaven and Potsdam, Germany. The stable water isotope ratios of ¹⁸O and D were measured by means of cavity ring-down spectroscopy (Picarro Inc. L-2120-i, and L-2140-i in AWI-Bremerhaven, L-2130-i in AWI-Potsdam.) and calibrated to the international VSMOW/VSLAP scale. Single measurements are provided with a precision of 0.1 and 1.0 % (1sd) for ¹⁸O and D. The accuracy of the measurement determined by a quality standard in each measurement run is 0,16 % for ¹⁸O and 1,1 for D %.

2.3 Data processing – Small-scale anomalies

The heavy stable water isotopes exhibit a depletion along the traverse with increasing altitude (Dansgaard, 1964) (Fig. 1 c). The observed relationship (-0.0037 % per meter of elevation, Fig. 1c) is consistent to the magnitude of trends expected from isotope enabled climate models within the studied area (ECHAM5-wiso model; Münch & Werner, 2020). Thus, to study the effect of local topography on water isotopic composition, we removed the observed linear trend of ¹⁸O against elevation along the route.

We derived the local heights (i.e. local topography) by first removing the longterm trend by subtracting a 20 km running mean and then reducing the noise and the small-scale variations from sastrugis by applying a 1 km running mean (Fig.2a). The local surface slope (Fig. 2b) is computed as the spatial derivative of the local heights. Similarly, local variations of accumulation rate (accumulation rate anomalies, Fig. 2) are derived as the residuals of the accumulation rate and its 20 km running mean.

3 Results and discussion

3.1 Small-scale spatial variability of accumulation rate and isotopic composition



Figure 2. Small-scale anomalies of accumulation rate related to local topography (a, b, d) and to isotopic composition anomalies (c, e).

The variation of local accumulation (red lines) is superposed to the local height of surface undulations (a, black line, left axis) and the surface slope (b, grey line, left axis reversed). Regression of local accumulation against local slope (d) (N=115, 100 m average every kilometer, standard-errors of 0.1 m km⁻¹ and 0.05 kg m⁻² yr⁻¹, respectively). The local topography at sampling sites is shown (a, yellow markers). The derived ¹⁸O anomalies along the route (c, left axis reversed, blue points connected by dashed line) are superposed to the local accumulation. Error bars in c) and d) indicate 1 standard error. Regression of ¹⁸O anomalies (N=27) against the 100 m average of the local accumulation at sites (e) (standard-error of 0.04 kg m⁻² yr⁻¹).

3.1.1 Relation of local accumulation rate and local surface topography

Extrema of local accumulation rate (maxima at kilometers 1.5, 36, 50, 64, 91, minima at kilometers 5, 40, 52, 77) surround the four distinct topographic features along the route (Fig. 2a) within a range of variation of 18.4 kg m⁻² yr⁻¹. With a prevailing wind in the area from North East to South West (Fujita et al., 2011), the deposition of drifted snow follows the pattern of increased and reduced accumulation rate in the hollows and bumps, a relationship that is well established (King et al., 2004).

Moreover, we find a strong relationship between the local surface slope and the accumulation rate anomalies ($\mathbb{R}^2 = 0.62$, p-value <0.001, Fig. 2b, Fig. 2d). This observation underlines the sensitivity of deposition to the slope of the surface (Fujita et al., 2011) as visible in the well-spread intervals of increased and reduced accumulation rate along the windward- and leeward-slopes of the surface, respectively (Arcone et al., 2005; Whillans, 1975).

3.1.2 Relation of local accumulation rate and isotopic composition of snow

The ¹⁸O anomalies along the traverse show strong variations (>2 ‰) that seem to be coupled to the local accumulation. Negative anomalies of ¹⁸O were generally found in regions with positive local accumulation rate anomalies and vice versa (Fig. 2c). A linear regression analysis (Fig. 2e) shows a statistically significant negative linear relationship ($R^2 = 0.31$, p-value 0.0029) between both variables. We note that the remaining stratigraphic noise on our isotopic anomalies likely explains the relatively low correlation and the true underlying relationship is likely stronger. Our finding implies that a change in accumulation rate of +10 kg m⁻² yr⁻¹ leads to a depletion of -0.52 ‰ in ¹⁸O (i.e. a sensitivity of -0.052 ‰ per kg m⁻² yr⁻¹). This sensitivity is furthermore fully consistent with the enrichment of ¹⁸O composition (+0.5 ‰ per kilometer) towards a reduced accumulation rate site (-8 kg m⁻² yr⁻¹) observed over a 5 km transect in the vicinity of Kohnen station (Dallmayr et al., 2020).

Importantly, the range of spatial variations in ¹⁸O is similar to the temporal variations observed in Antarctic ice-cores over the last millennium (Stenni et al., 2017) but has the opposite direction to the temporal climatic relationship between precipitation and temperature, with more precipitation linked to warmer temperatures (Frieler et al., 2015).

There are several hypotheses to explain this spatial relationship:

(a) Seasonal redistribution of snow: a larger redistribution during winter due to higher wind speeds (Birnbaum et al., 2010) and smaller snow crystal sizes would lead to more isotopically depleted winter summer snow in the hollows and enriched summer snow in the bumps, respectively. (Ekaykin et al., 2002).

(b) Post-depositional effects: as a consequence of the reduced accumulation, the surface snow is subject to longer and stronger post-depositional modifications, which are characterized by an enrichment in heavier isotopes as described by

Town et al. (2008). Evidence for this hypothesis was found by Ekaykin et al., (2016) interpreting the change in deuterium excess and 17 O. This idea is further supported by recent studies (Hughes et al., 2021; Ma et al., 2020a, 2020b).

While the data here doesn't allow to distinguish between both hypotheses, in either case the relationship of isotopes and local accumulation rate anomalies would have direct implications for the interpretation of ice-core records if the spatial changes also affect the downcore ice-core records. Ekaykin et al. (2016) suggested that a changing topography induced by moving megadunes together with an influence of topography on isotopes could create isotopic variations in ice-core records. While this only applies particularly to megadune areas, even stationary surface conditions (topography, accumulation rate, isotopic composition) will irremediably be advected by the ice flow over time and thus translate spatially varying surface conditions into temporal variations in ice-cores.

3.2 The B31-core: a spatial to temporal transfer induced by flowing ice

As an example, we study the B31 ice-core site. Here, the analysis of available radar profiles (Eisen et al., 2004; Rotchky et al., 2004) suggests that the topography of the study area was stationary over the last millennium, as constrained by bedrock topography (Steinhage et al., 1999). The ice sheet slowly flows towards South West at a velocity 3 m yr⁻¹ at the B31 site (satellite measurement, Fig. S1 in supporting information), and the topography upstream of the location exhibits a hollow with the lowest point ~2.5 km from the core-site (Fig. 3a). We independently estimated the ice-flow velocity from the GPR data (Arcone et al., 2005) by tracking the position of the lowest point over the last millennium in five dated internal reflection horizons (Eisen et al., 2004) (Fig. 3b). We find 3.8 m yr⁻¹ (±0.6 m yr⁻¹, 1sd) towards SW, consistent with the satellite data. Considering the uncertainties of both methods, we assume a velocity of 3.5 m yr⁻¹ to estimate the original position of the dated horizons (Fig. 3b). The results show that the recent horizons from the leeward slope >2.5 km upstream.

In the 1500 years old B31 record, (Oerter et al., 2004), the accumulation rate shows a steep increase around 1000AD followed by a stable accumulation rate with a broad local maximum at 1500AD (Fig. 3c). The ¹⁸O record shows the inverse behavior; a decrease of around 2.5 % from 500AD to 1000AD and more stable, slightly increasing values afterwards (Fig. 3d).



Figure 3. The B31 case: (a, b) upstream surface topography advected into the column at site, and records of (c) accumulation rate and (d) 18 O composition.

a) the surface origins related to the local heights. b) The 114 m long B31-core (red line) and the depth profiles of the five internal reflection horizons with their age estimate (Eisen et al., 2004). The dotted black line connects the nearest bottom of the upstream hollow (triangle) used for estimation of flow velocity. The black circles indicate the original surface positions of each age horizon along the core (red circles) assuming a constant flow of 3.5 m yr⁻¹ towards SW. c) 50 yr running mean of accumulation-rate record (orange line). The mean of the period 500-1000AD is marked with a circle, and the means of the nine 100 yr periods as squares. d) 50 yr running mean ¹⁸O record (blue line) and measurement of snow value of season 18/19 (blue marker). The predicted 100 yr ¹⁸O composition as induced by topography are shown by red squares (connected by dashed red line), with the mean of period 500-1000AD (red circle).

3.3 $^{18}\mathrm{O}$ record compared to the expected variations from the topographic effect

To predict the effect of the temporal ¹⁸O variations from the advection of the spatial variability, we assume that in first order, the accumulation rate at the site was constant over time. This assumption is supported by the analysis of B32 near Kohnen station that shows a nearly constant accumulation rate over time (Table S2, supporting information). This allows us to use the derived empirical relationship between isotopic anomalies and accumulation rate anomalies (i.e.

-0.052 ‰ per kg m⁻² yr⁻¹, section 3.1.2) together with the changes in accumulation rate to estimate the ¹⁸O anomalies induced by the topography ($\delta^{18}O_{\text{topo}}$)

$$\delta^{18}O_{\text{topo}}(t) = -0.052 * \Delta A (t - t_0) + \delta^{18}O(t_0) (1)$$

 $\Delta A(t-t_0)$ (in kg m⁻² yr⁻¹) refers to the accumulation rate anomaly between time (t) and the initial period (t_0) and ¹⁸O (t_0) is the initial isotopic composition. We choose 500–1000AD as the initial time period and perform the analysis on 100 yr mean values.

The resulting dataset is superposed to the 50 yr smoothed ¹⁸O temporal variations (Fig. 3c). The similar 1000-years trend between signals supports the hypothesis that this slow variation might be predominantly generated by the effect of topography, and not climate driven. This is further supported by comparing the B31 and B32 records over the last 1500 years (Fig. S2, supporting information). Being on the divide, the B32 site is not expected to be affected by the topography effect on millennial time-scales (as we show later, see Fig. 4) but due to their proximity (100km distance) we can assume similar climatic imprints. In contrast to B31, the isotope and accumulation time-series of B32 are rather constant but feature similar decadal to centennial variability, and the offset between both isotopic records can be explained by the difference in elevation between sites (~0.8 ‰ based on the depletion-elevation trend, Fig. 1c). This supports that the 1000-yrs variation observed in the B31 record are consistent to the expected upstream topographic anomaly.

4. Implications for other ice-cores and temperature reconstructions of the last millennium

To study how this interaction of spatial isotope variations linked to the local topography and ice-flow may affect other ice-core records, we derive a spatial map of the time scales that might be affected by this topographic effect. Assuming that a significant relationship between local topography and ¹⁸O anomalies also exists beyond our study area, and the ideal cases of stationary topography and constant ice-flow during the time period of interest, the affected time scales are the product of the length scale of surface undulations and the ice-flow velocity. We use the surface velocity distribution available across the continent (Mouginot et al., 2019) together with an estimated average 5 km length scale of surface undulation computed from Howat et al. (2019) (Text S2, supporting information).

The resulting map of the time scales potentially affected by the topographiceffect (Fig. 4) shows a wide range of affected timescales from several thousand years on the ice-divides down to centuries in regions with higher ice-flow. At the B32 site, the map predicts that the topographic effect does not act on millennial and faster time scales, whereas the map indicated time scale of around 1000 years at the B31 site, consistent with our detailed analysis.



Figure 4. Map of estimated time scale variations in the records due to the topographic effect induced by ice flow in Antarctica (left), and overview of East-Antarctica (right). Ice cores from plateau sites > 2000 m.a.s.l. used in the Antarctica 2k temperature reconstruction (Stenni et al., 2017) are indicated as red points, including the B31 site (white point). The grey contours show the 500 m elevation lines (Howat et al., 2019).

The core B31 was used in the community temperature reconstruction of the last millennium (Antarctica 2k) (Stenni et al., 2017) without special consideration of its non-climatic variability in the lowermost section. We can identify several additional cores included in the database located in similar ~1000 years scale variability estimation (red points in Fig. 4). For these cores, an analysis of the relationship between accumulation rate and isotopic records can provide further

insight, as a negative relation might be a signature of the topographic effect (section 3.1.2). In such case using the records for climate reconstruction should be avoided, carefully apprehended by an adequate upstream correction (beyond the scope of this study), or filtered by interpreting only the safe time scales. Finally, if the topographic effect that we propose here is confirmed by other studies, it should be also considered in the choice of future drilling sites.

5. Conclusions

We performed a 115 km traverse on the plateau of Dronning Maud Land with representative snow sampling (10 x 1 m snow liner per site) on 27 sites resolving the small-scale topographic variations combined with GPR and GPS. The high precision of the mean ¹⁸O snowpack values from averaging out stratigraphic noise allowed us to identify a significant negative relationship between the local anomalies of ¹⁸O and deposition driven by the topography ('topographic effect'). We propose and demonstrate on the example of the B31 site that iceflow advects these spatial anomalies into centennial to multi-millennial isotopic anomalies in ice cores and the topographic effect is thus highly relevant for the interpretation of ice-core. Confronting the estimated 'topographic effect' for the B31 core with the measured isotope record demonstrates that most of the millennial changes in this core can be explained without invoking climatic changes. As in many regions the topography on millennial scales is linked to the bedrock and thus stationary in time, we extrapolate our findings and provide a map of the potential topographic effect on ice-cores. This suggests that several icecores used in the reconstruction of the last millennium might be also affected by these non-climatic variations. Further studies on the physical origin of this relationship and its strength beyond our study region will help to refine our understanding, potentially allow for a correction of affected records and thus a more robust reconstruction of the Antarctic climate of the last millennia.

Data Availability

The GPR related data (traverse-00/01) is archived at https://doi.pangaea.de/1 0.1594/PANGAEA.935129 and https://doi.pangaea.de/10.1594/PANGAEA .935031, The traverse-18/19 data under https://doi.pangaea.de/10.1594/PAN GAEA.935030, https://doi.pangaea.de/10.1594/PANGAEA.935029.

PANGEA is hosted by the Alfred Wegener Institute Helmholtz Centre for Polar and Marine Research (AWI), Bremerhaven, and the Center for Marine Environmental Sciences (MARUM), Bremen, Germany.

Acknowledgment

This project was supported by Helmholtz funding through the Polar Regions and Coasts in the Changing Earth System (PACES) programme of the Alfred Wegener Institute. TL was supported by the European Research Council (ERC) under the European Union's Horizon 2020 Research and Innovation Programme (grant agreement no. 716092). We are grateful to the Kohnen 18/19 field season members, especially Klaus Trimborn for his decisive participation during the field traverse. We thank Olaf Eisen for providing the necessary data from the traverse 2000/01, and Thomas Münch for his input with the ECHAM5/wiso model.

Authors contribution

T.L. and M.H. enabled the project. T.L. M.H. and J.F. designed the traverse, performed the field-work, and provided support and guidance to the first author for the preparation of the manuscript. M.H. supervised this part of the PhD project. F.W. contributed to the discussion sections and edited the language of the paper. D.J. provided the Figure 4 using the geographic information system Q-GIS, and MB conducted all isotopic measurements at AWI-Bremerhaven. R.D. performed the field-work, analyzed the data and prepared this manuscript, reviewed by all co-authors.

References

Arcone, S., Spikes, V., & Hamilton, G. (2005). Stratigraphic variation within polar firn caused by differential accumulation and ice flow: Interpretation of a 400 MHz short-pulse radar profile from West Antarctica. Journal of Glaciology, 51(174), 407-422. doi:10.3189/172756505781829151

Birnbaum, G., Freitag, J., Brauner, R., König-Langlo, G., Schulz, E., Kipfstuhl, S., et al. (2010). Strong-wind events and their influence on the formation of snow dunes: Observations from Kohnen station, Dronning Maud Land, Antarctica. Journal of Glaciology, 56(199), 891-902. doi:10.3189/002214310794457272

Black, H., & Budd, W. (1964). Accumulation in the Region of Wilkes, Wilkes Land, Antarctica. Journal of Glaciology, 5(37), 3-15. doi:10.3189/S0022143000028549

Casado, M., Landais, A., Picard, G., Münch, T., Laepple, T., Stenni, B., et al. (2018). Archival processes of the water stable isotope signal in East Antarctic ice cores. The Cryosphere, 12, 1745–1766, doi:10.5194/tc-12-1745-2018.

Casado, M., Münch, T., and Laepple, T. (2020). Climatic information archived in ice cores: impact of intermittency and diffusion on the recorded isotopic signal in Antarctica, Climate of the Past, 16, 1581–1598. doi.org/10.5194/cp-16-1581-2020

Dallmayr, R., Freitag, J., Hörhold, M., Laepple, T., Lemburg, J., Della-Lunga, D., & Wilhelms, F. (2021). A dual-tube sampling technique for snowpack studies. Journal of Glaciology, 67(261), 84-90. doi:10.1017/jog.2020.85

Dansgaard, W. (1964). Stable isotopes in precipitation, Tellus, 16:4, 436-468, doi 10.3402/tellusa.v16i4.8993

Eisen, O., Nixdorf, U., Wilhelms, F., & Miller, H. (2004). Age estimates of isochronous reflection horizons by combining ice core, survey, and synthetic radar data. Journal of Geophysical Research, 109(B4), B04106. doi:10.1029/2003JB002858

Eisen, O., Rack, W., Nixdorf, U., & Wilhelms, F. (2005). Characteristics of accumulation around the EPICA deep-drilling site in Dronning Maud Land, Antarctica. Annals of Glaciology, 41, 41-46. doi:10.3189/172756405781813276

Ekaykin, A., Lipenkov, V., Barkov, N., Petit, J., & Masson-Delmotte, V. (2002). Spatial and temporal variability in isotope composition of recent snow in the vicinity of Vostok station, Antarctica: Implications for ice-core record interpretation. Annals of Glaciology, 35, 181-186. doi:10.3189/172756402781816726

Ekaykin, A., Eberlein, L., Lipenkov, V., Popov, S., Scheinert, M., Schröder, L., & Turkeev, A. (2016). Non-climatic signal in ice core records: lessons from Antarctic megadunes. The Cryosphere, 10, 1217–1227, doi:10.5194/tc-10-1217-2016

Fahnestock, M. A., Scambos, T. A., Shuman, C., Arthern, R. J., Winebrenner, D. P., & Kwok, R. (2000). Snow megadune fields on the East Antarctic Plateau: Extreme atmosphere-ice interaction, Geophysical Research Letters, 27(20), 3719–3722. doi:10.1029/1999GL011248

Fisher, D., Reeh, N., & Clausen, H. (1985). Stratigraphic noise in time series derived from ice cores. Annals of Glaciology, 7, 76–83. doi:10.3189/S0260305500005942

Frezzotti, M., Gandolfi, S., & Urbini, S. (2002). Snow megadunes in Antarctica: Sedimentary structure and genesis. J. Geophys. Res., 107(D18), 4344, doi:10.1029/2001JD000673

Frezzotti, M., Pourchet, M., Flora, O., Gandolfi, S., Gay, M., Urbini, S., et al. (2004). New estimations of precipitation and surface sublimation in East Antarctica from snow accumulation measurements. Climate Dynamics, 23, 803–813, doi:10.1007/s00382-004-0462-5

Frieler, K., Clark, P., He, F., Buizert, C., Reese, R., Ligtenberg, S.R.M., et al. (2015). Consistent evidence of increasing Antarctic accumulation with warming. Nature Climate Change, 5, 348–352. doi:10.1038/nclimate2574

Fujita, S., Holmlund, P., Andersson, I., Brown, I., Enomoto, H., Fujii, Y., et al. (2011). Spatial and temporal variability of snow accumulation rate on the East Antarctic ice divide between Dome Fuji and EPICA DML. The Cryosphere, 5, 1057–1081, doi:10.5194/tc-5-1057-2011

Howat, I. M., Porter, C., Smith, B. E., Noh, M.-J., & Morin, P. (2019). The Reference Elevation Model of Antarctica. The Cryosphere, 13, 665–674, doi:10.5194/tc-13-665-2019

Hughes, A. G., Wahl, S., Jones, T. R., Zuhr, A., Hörhold, M., White, J. W. C., & Steen-Larsen, H. C. (2021). The role of sublimation as a driver of climate signals in the water isotope content of surface snow: Laboratory and field experimental results. The Cryosphere Discuss. doi:10.5194/tc-2021-87.

Jones, J.M., Gille, S.T., Goosse, H., Abram, N.J., Canziani, P.O., Charman, D.J. et al. (2016). Assessing recent trends in high-latitude Southern Hemisphere surface climate. Nature Climate Change 6, 917–926 (2016). doi.org/10.1038/nclimate3103

Jouzel J., Alley, R. B., Cuffey, K. M., Dansgaard, W., Grootes, P., Hoffmann, et al. (1997). Validity of the temperature reconstruction from water isotopes in ice cores. Journal of Geophysical Research 102(C12), 471–487 doi: 10.1029/97JC01283.

King, J. C., Anderson, P. S., Vaughan, D. G., Mann, G. W., Mobbs, S. D., & Vosper, S. B. (2004). Wind-borne redistribution of snow across an Antarctic ice rise. Journal of Geophysical Research, 109, D11104, doi:10.1029/2003JD004361.

Klein, F., Abram, N. J., Curran, M. A. J., Goosse, H., Goursaud, S., Masson-Delmotte, et al. (2019). Assessing the robustness of Antarctic temperature reconstructions over the past 2 millennia using pseudoproxy and data assimilation experiments. Climate of the Past, 15, 661–684, doi: 10.5194/cp-15-661-2019.

Laepple, T., Hörhold, M., Münch, T., Freitag, J., Wegner, A., Kipfstuhl, S. (2016). Layering of surface snow and firn at Kohnen Station, Antarctica: Noise or seasonal signal?. Journal of Geophysical Research Earth Surface, 121, 1849–1860, doi:10.1002/2016JF003919

Ma, T., Li, L., Li, Y., An, C., Yu, J, Ma, H., et al. Jiang, S., Shi, G. (2020). Stable isotopic composition in snowpack along the traverse from a coastal location to Dome A (East Antarctica): Results from observations and numerical modeling. Polar Science, 24, 1873-9652, doi: 10.1016/j.polar.2020.100510.

Ma T., Li L., Shi G., & Li Y. (2020). Acquisition of Post-Depositional Effects on Stable Isotopes (180 and D) of Snow and Firn at Dome A, East Antarctica. Water., 12(6):1707. doi:10.3390/w12061707

Medley, B., McConnell, J. R., Neumann, T. A., Reijmer, C. H., Chellman, N., Sigl, M., & Kipfstuhl, S. (2018). Temperature and snowfall in western Queen Maud Land increasing faster than climate model projections. Geophysical Research Letters, 45, 1472–1480. doi:10.1002/2017GL075992

Mouginot, J., Rignot, E., & Scheuchl B. (2019). Continent-wide, interferometric SAR phase-mapping of Antarctic ice velocity. Geophysical Research Letters, 46, 9710-9718. doi:10.1029/2019GL083826

Münch T, Kipfstuhl S, Freitag J, Meyer H & Laepple T. (2016). Regional climate signal vs. Local noise: a two-dimensional view of water isotopes in Antarctic firn at Kohnen station, Dronning Maud Land. Climate of the Past, 12, 1565–1581. doi:10.5194/cp-12-1565-2016

Münch, T., Kipfstuhl, S., Freitag, J., Meyer, H., and Laepple, T. (2017). Constraints on post-depositional isotope modifications in East Antarctic firm from analyzing temporal changes of isotope profiles, The Cryosphere, 11, 2175–2188. doi:10.5194/tc-2017-35.

Münch, T., &Laepple, T. (2018). What climate signal is contained in decadalto centennial-scale isotope variations from Antarctic ice cores? Climate of the Past, 14, 2053–2070, doi: 10.5194/cp-14-2053-2018

Münch T., & Werner, M. (2020). Antarctic time series of temperature, precipitation, and stable isotopes in precipitation from the ECHAM5/MPI-OM-wiso past1000 climate model simulation. DOT: 10.5281/zenodo.4001565

Neumann, T. A., & Waddington, E. D. (2004). Effects of firm ventilation on isotopic exchange. Journal of Glaciology, 50 (169), 183–194, doi:10.3189/172756504781830150

Oerter, H., Graf, W., Meyer, H., & Wilhelms, F. (2004). The EPICA ice core from Dronning Maud Land: First results from stable-isotope measurements. Annals of Glaciology, 39, 307-312. doi:10.3189/172756404781814032

PAGES 2k Consortium. (2013). Continental-scale temperature variability during the past two millennia, Nature Geoscience, 6, 339–346, Published online 21 April 2013, doi:10.1038/NGEO1797

Rotschky, G., Eisen, O., Wilhelms, F., Nixdorf, U., & Oerter, H. (2004). Spatial distribution of surface mass balance on Amundsenisen plateau, Antarctica, derived from ice-penetrating radar studies. Annals of Glaciology, 39, 265-270. doi:10.3189/172756404781814618

Rotschky, G., Holmlund, P., Isaksson, E., Mulvaney, R., Oerter, H., Van Den Broeke, M., & Winther, J. (2007). A new surface accumulation map for western Dronning Maud Land, Antarctica, from interpolation of point measurements. Journal of Glaciology, 53(182), 385-398. doi:10.3189/002214307783258459

Steinhage, D., Nixdorf, U., Meyer, U., & Miller, H. (1999). New maps of the ice thickness and subglacial topography in Dronning Maud Land, Antarctica, determined by means of airborne radio-echo sounding. Annals of Glaciology, 29, 267-272. doi:10.3189/172756499781821409

Steinhage, D., Kipfstuhl, S., Nixdorf, U., & Miller, H. (2013). Internal structure of the ice sheet between Kohnen station and Dome Fuji, Antarctica, revealed by airborne radio-echo sounding. Annals of Glaciology, 54(64), 163-167. doi:10.3189/2013AoG64A113

Stenni, B., Curran, M. A. J., Abram, N. J., Orsi, A., Goursaud, S., Masson-Delmotte, V., et al. (2017). Antarctic climate variability on regional and continental scales over the last 2000 years, Climate of the Past, 13, 1609–1634, doi:10.5194/cp-13-1609-2017

Steig, E., Ding, Q., White, J. W.C., Küttel, M., Rupper, S. B., Neumann, T. A., et al. (2013). Recent climate and ice-sheet changes in West Antarctica compared with the past 2,000 years. Nature Geoscience, 6, 372–375. doi: 10.1038/ngeo1778

Town, M. S., Warren, S. G., Walden, V. P., & Waddington, E. D. (2008).

Effect of atmospheric water vapor on modification of stable isotopes in near-surface snow on ice sheets. Journal of Geophysical Research, 113, D24303, doi:10.1029/2008JD009852

Whillans, I. (1975). Effect of Inversion Winds on Topographic Detail and Mass Balance on Inland Ice Sheets. Journal of Glaciology, 14(70), 85-90. doi:10.3189/S0022143000013423